Author’s response to reviews

Title: Pimping: A tradition of gendered disempowerment

Authors:

David Chen (chend4@uw.edu)
Kelsey Priest (priest@ohsu.edu)

Version: 2 Date: 18 Aug 2019

Author’s response to reviews:

1. Regarding abbreviations, NASEM was the only one that had been used. However, since it was only used once, I reasoned that an abbreviation was not necessary, and I removed it.

2. Regarding funding, though Kelsey does receive funding for her research, none of that was directed to or applied to this manuscript. I therefore put, ”Not applicable,” for funding.

3. Regarding responding to the reviewers’ comments, no response was warranted.